

Responses to Reviewers' Comments

☐ **Manuscript ID:** <Manuscript ID>

☐ **Title:** <Manuscript Title>

☐ **Author:** <Author Names>

To Editor and Reviewers:

<Your response to the editor and reviewers, including a brief summary of the major revisions made to the manuscript.>

Reviewer 1

Summary Comment

<Paste the reviewer summary comment here>

Response:

<Write your high-level thanks and summary response here>

Comment 1-1:

<Reviewer comment 1>

Response:

<Your response to comment 1>

Changes:

<What changed in manuscript, include section/page/equation/table if possible>

Comment 1-2:

<Reviewer comment 2>

Response:

<Your response to comment 2>

Changes:

<What changed in manuscript for comment 2>

